**Author’s response to reviews**

**Title:** Exempting Low-Risk Health and Medical Research from Ethics Reviews: comparing Australia, United Kingdom, United States and The Netherlands

**Authors:**

Anna Scott (ascott@bond.edu.au)

Simon Kolstoe (simon.kolstoe@port.ac.uk)

M.C. Ploem (m.c.ploem@amsterdamumc.nl)

Zoë Hammatt (zhhconsulting@gmail.com)

Paul Glasziou (pglaszio@bond.edu.au)

**Version:** 2 **Date:** 03 Dec 2019

**Author’s response to reviews:**

Responses to peer reviewer comments are provided in an uploaded document called "Scott et al - responses to peer review."